Author's response to reviews

Title: Audio computer-assisted self-interviewing (ACASI) may avert socially desirable responses about infant feeding in the context of HIV

Authors:

Anthony K Waruru (awaru@kisian.mimcom.net)
Ruth Nduati (rduati@icconnect.co.ke)
Thorkild Tylleskar (thorkild.tylleskar@cih.uib.no)

Version: 2 Date: 23 November 2004

Author's response to reviews:

Kisumu Nov 23, 2004

The Editor, Journal of Medical Informatics and Decision Making, Biomed Central,

Dear Editors,

Re: Changes on the manuscript for the Journal of Medical Informatics and Decision Making
Changes have been made describing the contents of figure 2 within the main body (page 8) as follows. A graphical presentation of 3 of the 13 food items: milk, eggs and herbal tea is presented in (figure 2). Previously, the sentence read 'A graphical presentation of 2 of the 13 food items: milk, and protein-rich foods (eggs, meat or fish) is presented in (figure 2).'